Dear Editor,

Thank you for carefully reviewing our manuscript previously titled “iMCD and TAFRO syndrome: to improve the diagnostic consciousness is the key” for possible publication in the World Journal of Clinical Cases. We are grateful to you and your reviewers for their constructive critique. We have revised the manuscript, highlighting our revisions in red. and have attached point-by-point responses detailing how we have revised the manuscript in response to the reviewers' comments below.

Thank you for your consideration and further review of our manuscript. Please do not hesitate to contact us with any further questions or recommendations.

Yours Sincerely,
Qian-Yun Zhou
2021-10-28
Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: The authors focus on a rare hematological disease with a high impact on individual clinical status, namely, the generalized compromise in multicentric form. the article is an observational and retrospective study
Response: Thanks.

Reviewer #2:
1. Your abstract needs to be improved. Then the second sentence should start "We report the case of..." giving a very brief description. A third sentence should describe in general terms the treatment and the outcome.
Response: This study is a retrospective case analysis, not a case report.
2. There is no clinical image attached.
Response: This study is a retrospective case analysis, not a case report. There are no clinical images to be involved in the article.
3. Rest is good and nicely described in discussion portion.
Response: Thanks.